Author's response to reviews

Title: Council tax valuation bands, socio-economic status and health status: a cross-sectional analysis from the Caerphilly Health and Social Needs Study

Authors:

David L Fone (foned@cf.ac.uk)
Frank Dunstan (dunstanfd@cf.ac.uk)
Stephen Christie (stephen.christie012@msd.govt.nz)
Andrew Jones (andrew.jones@nphs.wales.nhs.uk)
Jonathan West (jonathan.west@nphs.wales.nhs.uk)
Margaret Webber (margaret.webber@nphs.wales.nhs.uk)
Nathan Lester (nathan.lester@nphs.wales.nhs.uk)
John Watkins (john.watkins@nphs.wales.nhs.uk)

Version: 3 Date: 20 April 2006

Author's response to reviews:

Abstract- Please shorten the abstract to no more than 350 words. Please remember to also update the abstract details on the submission system.

I confirm that I have shortened the abstract to 345 words and updated on the submission system

Methods- Please document in the Methods section the type of consent sought from the participants.

I have added an appropriate sentence.

Competing interests - If there are none to declare, please write 'The authors declare that they have no competing interests'.

I have included this sentence

Figures - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping.

I have done my best